Author's response to reviews

Title: In-vitro and in-vivo validation of ethnopharmacological uses of Isodon rugosus

Authors:

Khalid Hussain Janbaz (khjanbaz@hotmail.com)
Javeria Arif (javeria.arif46@yahoo.com)
Fatima Saqib (fatimasaqib786@yahoo.com)
Imran Imran (imran.ch@bzu.edu.pk)
Muhammad Ashraf (muhammadashraf291@yahoo.com)
Muhammad Zia-Ul-Haq (ahirzia@gmail.com)
Hawa Z.E. Jaafar (hawazej@gmail.com)
Vincenzo De Feo (defeo@unisa.it)

Version: 2 Date: 9 October 2013

Author's response to reviews: see over
Dear Editor

BMC Complementary and Alternative Medicine

Subject: submission of revised ms no. 8769346731097018

Dear Sir

We are highly thankful for your kind consideration of manuscript.

Following are answers of comments raised

1. We have added ethical statement approval of animals for this study. Also please find attached scanned notification of same.

2. We have added previous animal studies references. Although kit methods are available for some bioassays thus bypassing animal studies, however such kits are not available for smooth muscle tissue activity, major part of our paper.

I hope you will consider our points and will process our manuscript.

Thank you.

I am available if there are any further queries.

-----

Highest Regards

Prof. Dr. Vincenzo De Feo

Dipartimento di Scienze Farmaceutiche e Biomediche,

Università degli Studi di Salerno, Italy

E-mail: defeo@unisa.it